Author's response to reviews

Title: Antisense oligonucleotides and all-trans retinoic acid have a synergistic anti-tumor effect on oral squamous cell carcinoma

Authors:

Qin Xu (xuqin_2004@hotmail.com)
Zhiyuan Zhang (Zhangzhiyuan1951@hotmail.com)
Ping Zhang (pingzhang71@hotmail.com)
Wantao Chen (chenwantao2002@hotmail.com)

Version: 6 Date: 7 May 2008

Author's response to reviews: see over
Dear Editor:

We appreciate you for publishing our manuscript (1813594549174437) entitled “Antisense oligonucleotides and all-trans retinoic acid have a synergistic anti-tumor effect on oral squamous cell carcinoma”.

We appreciate the points made by the editor. Accordingly, we checked carefully and revised the manuscript and provide below point-by-point responses regarding your suggestions.

1. We changed the formatting of italicise or bold text except headings and title.
2. We used initials to refer to each author's contribution.
3. All the figures were cropped as closely as possible to minimise white space around the image.
4. We corrected the formatting of referens.

We hope that our manuscript will be published as early as it could. Please let me know if we can provide further information.

Sincerely yours,

Wantao Chen
Ninth People’s Hospital
Shanghai JiaoTong University
639, Zhizaoju Road
Shanghai, 200011, China
Tel: 86-21-63138341-5211
Fax: 86-21-63135412
Email: chenwantao2002@hotmail.com